PERSPECTIVE

For reprint orders, please contact: reprints@futuremedicine.com

A Delphi consensus and open debate on the role of first-line bevacizumab for HER2-negative metastatic breast cancer

Fabio Puglisi^{*,1}, Giancarlo Bisagni², Mariangela Ciccarese³, Caterina Fontanella¹, Teresa Gamucci⁴, Luigi Leo⁵, Annamaria Molino⁶, Rosa Rita Silva⁷ & Paolo Marchetti⁸

To gain consensus on the role of bevacizumab plus paclitaxel as first-line treatment for HER2-negative metastatic breast cancer, a panel of expert oncologists experienced in treating patients with metastatic breast cancer in Italy participated in a Delphi consensus study. The panel reached a full consensus on the efficacy of bevacizumab plus paclitaxel and the clinical meaningfulness of the progression-free survival benefit compared with paclitaxel alone, despite the lack of an overall survival effect in clinical trials. The participants agreed that real-world data support the effectiveness and well-defined safety profile of the regimen. Views on the use of bevacizumab plus paclitaxel in specific patient populations were not unanimous and clinical judgment remains important. Nevertheless, a high level of agreement was reached.

First draft submitted: 22 June 2016; Accepted for publication: 8 July 2016; Published online: 22 July 2016

In recent years, there has been controversy about the antiangiogenic agent bevacizumab as treatment for HER2-negative metastatic breast cancer (mBC) [1-4]. After initial accelerated approval by the US FDA, based on the results of the E2100 trial [5], regulatory approval was withdrawn in the USA. In Europe, however, regulatory authorities examined the same clinical trial results and reached a different conclusion. The European regulatory approval of bevacizumab plus paclitaxel was retained and indeed expanded to include the combination of bevacizumab plus capecitabine (based on results of the RIBBON-1 trial [6]), although approval of bevacizumab plus docetaxel, as investigated in the AVADO trial [7], was withdrawn. The controversy continued when randomized trials in the neoadjuvant setting consistently demonstrated a benefit from the addition of bevacizumab to preoperative chemotherapy for early breast cancer [8-12], although effects on disease-free survival and overall survival (OS) showed less homogeneous results [13-15].

Differences of opinions from regulatory authorities are also reflected in the medical community. In the complex treatment decision-making process, varying opinions on the scientific interpretation of end points, significance, applicability and clinical meaningfulness of results all play important roles in the overall decision to use – or not use – bevacizumab to treat a patient with mBC. To

¹Department of Medical & Biological Sciences, University of Udine, Udine, Italy

²Oncology Unit, Department of Oncology, Azienda Ospedaliera ASMN, Istituto di Ricovero e Cura a Carattere Scientifico, Reggio Emilia, Italy

Ĩo

KEYWORDS

angiogenesis

cancer

bevacizumab
 breast

Future

NCOLOG



³Department of Medical Oncology, Vito Fazzi Hospital, Lecce, Italy

⁴Medical Oncology Unit, ASL Frosinone, Frosinone, Italy

⁵Oncology Unit, Azienda Ospedaliera dei Colli, Naples, Italy

⁶University of Verona, Verona, Italy

⁷Medical Oncology Unit, ASUR Marche AV2 Fabriano, Fabriano, Italy

⁸Medical Oncology, Sant'Andrea Hospital, Sapienza University of Rome & IDI-IRCCS, Rome, Italy

^{*}Author for correspondence: fabio.puglisi@uniud.it

gain insight into the scientific rationale for such decisions, a Delphi consensus method was used to evaluate the degree of agreement and disagreement about the possible therapeutic role of bevacizumab in HER2-negative mBC. In challenging situations where scientific evidence is lacking or conflicting, the application of the Delphi consensus method enables a panel of independent experts to reach the most unanimous opinion possible on a given topic [16–19].

In this article, we present the statements defined in the consensus and the data supporting the conclusions drawn, but also expand on this by discussing some of the most recently published and presented data, as well as speculating on future directions and new and emerging strategies.

Method

The Delphi consensus model implemented in this study was developed over a period of 2 months and consisted of three key steps. In the first step, a scientific board comprising eight experts in mBC was convened to define the statements on which a broader panel would vote in later steps. The members of the scientific board were identified based on their publication record and participation in national and international medical congresses and meetings. The scientific board met in Rome on 2 April 2015, joined by a psychologist with expertise in qualitative research in the pharmaceutical sector, who moderated the meeting. During this session, the panelists discussed potential topics for future online voting and ultimately defined 12 statements related to the appropriateness and therapeutic role of the combination of bevacizumab plus paclitaxel in various populations of patients with HER2-negative mBC.

In the second step, the statements agreed upon in step 1 were distributed to 37 oncologists with expertise in breast cancer. These comprised 31 oncology specialists identified by the members of the scientific board based on their clinical experience in breast cancer and six members of the scientific board. The two remaining members of the scientific board were excluded from the vote as they were chosen to chair the third step described below.

Of the 37 oncologists invited to take part in the online survey, 31 (84%) accepted. The survey was administered using a commercial software program (SurveyMonkey[®]), which has been successfully used in other Delphi consensus surveys [16,20]. Using this platform, the 12 statements, along with results of supporting studies and publications from the literature, were sent to the 31 participating panelists on 7 May 2015. Panel members were asked to reply within 11 days. Panelists were asked, individually and anonymously, to express their level of agreement with each statement using a fiveitem Likert scale (where 1 = completely disagree; 2 = slightly disagree; 3 = partially agree; 4 = agree; 5 = completely agree) and the main reason for their chosen level of agreement from a series of prespecified options. A consensus was considered to have been reached if the sum of answers 1 and 2 (negative) or 3, 4 and 5 (positive) exceeded 66%, as described previously for similar projects using the Delphi method [19,21-22]. The survey results were collected and the final report was shared with all participants on 19 May 2015.

The third step (the Delphi plenary session) was conducted during two web conferences on 21 May and 26 May 2015, using the WebEx® system. All survey participants were invited to discuss the results shared previously and to vote again on those statements for which the criterion for consensus (at least 66% unanimity) had not been reached in step 2. It was decided to organize two separate web meetings according to geographical area (northern Italy chaired by F Puglisi and southern Italy chaired by P Marchetti) and the availability of the specialists, but most importantly to ensure that the number of participants did not prevent full engagement in the discussion and the opportunity to voice opinions. Members of the scientific board and the psychologist took part in both web conferences. Overall, 27 of the 31 panelists participated in the two web conferences.

Results

After the online survey in the second step, the predefined threshold for consensus (at least 66% unanimity) was reached for all statements and therefore all statements were approved (Table 1). Because of the high level of agreement in step 2 (more than 90% consensus according to the prespecified criteria for all 12 statements), the two web conferences were used instead to discuss the results and the reasons for the answers given.

• Efficacy of first-line bevacizumab plus paclitaxel in HER2-negative mBC

1. Bevacizumab plus paclitaxel is an effective first-line option for HER2-negative mBC

In Italy, bevacizumab is indicated in combination with paclitaxel for the first-line treatment of

ltem	Statement	Degree of
item	Statement	consensus
First-line efficacy of the combination of bevacizumab plus paclitaxel in HER2-negative mBC	1. Bevacizumab plus paclitaxel is an effective first-line option for HER2-negative mBC	100%
	2. The advantage in terms of PFS achieved with bevacizumab plus paclitaxel in the first-line treatment of HER2-negative mBC is clinically significant	100%
	3. The lack of a benefit in terms of OS of bevacizumab as first-line treatment for mBC does not justify not using the drug	100%
Special populations	4. In the first-line treatment of HER2-negative mBC, the bevacizumab plus paclitaxel combination is a valid treatment option for patients with high disease burden	97%
	5. Bevacizumab plus paclitaxel is an adequate first-line treatment option for TNBC	97%
	6. Bevacizumab plus paclitaxel is an appropriate first-line treatment option for hormone receptor (ER/PgR)-positive HER2-negative mBC	100%
	7. In elderly patients, the use of bevacizumab is efficacious and safe as first-line treatment for HER2-negative mBC	97%
	8. Bevacizumab is a valid treatment option also in patients pretreated with (neo)adjuvant taxanes in the early BC setting	91%
Real life	9. In the mBC setting, the data obtained in real-life studies confirm the efficacy of bevacizumab obtained in randomized clinical studies	100%
Treatment duration	10. It is useful to continue the first-line treatment of mBC with bevacizumab until progression even after the discontinuation of chemotherapy	94%
QoL and safety	11. Combining bevacizumab with paclitaxel as first-line treatment for mBC does not negatively impact on QoL	97%
	12. Bevacizumab has a well-defined safety profile	100%

adult patients with HER2-negative mBC at the recommended dose of 10 mg/kg every 2 weeks or 15 mg/kg every 3 weeks [23]. There was unanimous agreement that bevacizumab plus paclitaxel is an effective first-line option in HER2-negative mBC, supported by the results of two randomized Phase III trials (the open-label E2100 trial [5,24] and, more recently, the double-blind placebo-controlled MERiDiAN trial [25]). Both trials showed that adding bevacizumab to firstline paclitaxel significantly improves progression-free survival (PFS; primary end point) and also overall response rate (ORR; secondary end point). The median PFS in both of these trials was approximately 11 months, although the hazard ratios (HRs) differed between the two trials. Of note, three additional randomized Phase III trials - TURANDOT, CALGB 40502 and TABEA - demonstrated almost identical median PFS durations with bevacizumab plus paclitaxel in this setting (11.0 months in TURANDOT, 11.0 months in CALGB 40502 and 11.3 months in TABEA) [26-28]. Similarly, ORRs with this regimen are very consistent (50% in E2100, 54% in MERiDiAN, 44% in TURANDOT, 38% in CALGB 40502 and 47% in TABEA), taking into account slight differences in trial design, study populations and tumor assessments.

PFS was the primary end point in all of these trials and its value is discussed in more detail below. However, in our opinion, ORR is also a very meaningful important end point in mBC. Responding patients have a better OS expectancy than those with stable or progressive disease, as shown in two large meta-analyses each including more than 2000 patients with mBC [29,30]. Furthermore, ORR is particularly important from the patient's perspective. Patients are reassured to know that their disease is responding and their lesions are shrinking during treatment. If a patient is responding to a given treatment, they can be advised that it is worth continuing with that regimen because their disease is under control. It is not difficult to imagine the psychological impact that such news will have on quality of life (QoL) and outlook for patients.

The statement discussed by the panelists relates to the combination of bevacizumab plus paclitaxel. It should also be noted that the double-blind randomized Phase III RIBBON-1 trial demonstrated significantly improved PFS (primary end point) with the addition of bevacizumab to capecitabine [6] and this regimen was compared with bevacizumab plus paclitaxel in the TURANDOT randomized Phase III trial mentioned above [26]. However, as the bevacizumab plus capecitabine regimen is not available in Italy, the discussion focuses on the bevacizumab plus paclitaxel combination.

2. The advantage in terms of PFS achieved with bevacizumab plus paclitaxel in the first-line treatment of HER2-negative mBC is clinically significant

The Phase III E2100 and MERiDiAN trials mentioned above demonstrated statistically significant improvements in PFS with the addition of bevacizumab to paclitaxel. The clinical significance of the results represents a more subjective question. In E2100, the HR according to Independent Review Facility assessment was 0.48 [24], representing a 52% reduction in the risk of progression or death. In MERiDiAN the HR was 0.68 (32% reduction in the risk of progression or death). There was little doubt when the results from the E2100 trial were first presented that halving the risk of reduction of PFS and median PFS values of 11.3 months with bevacizumab plus paclitaxel versus 5.8 months with paclitaxel alone were clinically significant, important and relevant. More modest differences were observed in subsequent trials, not only in the MERiDiAN trial [25] but also in the AVADO [7] and RIBBON-1 [6] trials evaluating bevacizumab in combination with alternative chemotherapy backbones. These trials had the methodological advantage of being doubleblind placebo-controlled trials whereas E2100 was an open-label trial. However, in E2100 an Independent Review Facility reported almost identical PFS results to the investigator-assessed PFS results [24].

There are at least three aspects to consider when assessing statement 2: first, is PFS a clinically significant (or relevant) end point? Second, what magnitude of difference is clinically significant? And third, what PFS improvement, if any, matters to a patient? In the authors' opinion, PFS is a clinically significant end point. From the patient's perspective, PFS (and ORR) are valuable (as mentioned elsewhere in this article). Regarding the relevance of this end point in formally defining treatment efficacy in prospective interventional clinical trials, we believe that PFS is more valuable than OS, interpretation of which can be confounded by postprogression survival and second and subsequent lines of therapy (see statement 3 below). The second question is also quite subjective, and an American Society of Clinical Oncology Cancer Research

Committee convened specifically to examine and define 'clinical meaningfulness' could not reach a consensus in breast cancer [31]. For patients with triple-negative mBC, an improvement in median PFS of 4 months was considered clinically meaningful [31]. The European Society for Medical Oncology Magnitude of Clinical Benefit Scale also attempts to define 'clinically meaningful' [32]. Specifically in trials with a primary end point of PFS, an HR of ≤ 0.65 together with a median PFS gain of \geq 1.5 months (if the control arm median is ≤ 6 months) or of ≥ 3 months (if the control arm median is >6 months) meets the criteria for the highest grade of benefit. Using these thresholds, E2100 but not MERiDiAN would be considered clinically meaningful, although it is important to note that the European Society for Medical Oncology Magnitude of Clinical Benefit Scale considers four additional aspects and should not be used to consider PFS in isolation. Moving beyond these theoretical and mathematical models, what are the views of a patient? In a survey of almost 300 women with mBC, 'extending PFS' was ranked the second most important treatment outcome after OS [33]. Importantly, results of the survey suggested that from a patient perspective, extending PFS is associated with improvement in QoL, physical functioning and emotional well-being.

3. The lack of a benefit in terms of OS of bevacizumab as first-line treatment for mBC does not justify not using the drug

OS is frequently cited as the 'gold standard' end point in trials of mBC [34,35]. However, in the first-line treatment setting in HER2-negative mBC, few if any trials have demonstrated a statistically significant improvement in OS and the factors contributing to this lack of OS difference, particularly postprogression survival and crossover, have been discussed intensely in the literature [35–38]. Although some postulate that antiangiogenic strategies may lead to a more aggressive tumor phenotype, evidence to support this hypothesis is limited to preclinical studies of tyrosine kinase inhibitors [39]. Furthermore, analyses of clinical trials of bevacizumab did not support these preclinical observations [40].

Although no evidence of OS benefit was observed in first-line clinical trials in HER2negative mBC, a gradual improvement in OS has been reported in mBC [41] and the most likely explanations, beyond improved screening, diagnosis and locoregional therapy, seem to be the incremental benefits in PFS with new therapies and the increasing number of regimens available to patients, enabling multiple lines of therapy. Thus, even if one specific treatment cannot be shown to significantly improve OS, the cumulative contribution of several different agents over the course of several lines of treatment can reasonably be expected to make a difference to OS. Therefore in the view of the panelists, it is difficult to exclude any particular therapy because of its lack of OS benefit demonstrated during use in a single treatment line. Indeed, if only treatments with proven OS benefit in contemporary clinical trials are to be used in the first-line setting of HER2-negative mBC, the options become extremely limited. Interestingly, there is a similar lack of evidence for a significant OS benefit in trials evaluating second and further lines of treatment [42].

Special populations

4. In the first-line treatment of HER2-negative mBC, the bevacizumab plus paclitaxel combination is a valid treatment option for patients with high disease burden

One of the major challenges when selecting patients for bevacizumab-containing therapy is the lack of a predictive marker, or indeed a clinical marker, identifying patients most or least likely to derive benefit from bevacizumab. An extensive biomarker program has been undertaken, exploring a range of tumor markers [43]. However, despite initially encouraging signals for the utility of plasma VEGF-A as a predictive biomarker for bevacizumab efficacy, recently presented results of the MERiDiAN randomized Phase III trial, which was designed to evaluate this candidate biomarker prospectively, did not support use of baseline plasma VEGF-A to identify patients benefiting most from bevacizumab [25].

In the absence of a robust and reliable predictive biomarker for bevacizumab efficacy, we tend to resort to clinical parameters to try to identify those patients who may be most suitable for bevacizumab therapy. However, numerous retrospective, *post hoc*, and exploratory analyses provide no clear evidence that one subgroup of patients benefits more (or less) from bevacizumab than another [44]. In this scenario, arguably there is no reason to favor use of bevacizumab in one population versus another. However, in a healthcare system with limited resources, we may often be faced with the challenge of limiting a treatment to a smaller number of patients than those strictly eligible for the treatment. Thus, although bevacizumab plus paclitaxel is approved in Europe for all patients with HER2-negative mBC unless they have a contraindication for bevacizumab, reimbursement constraints pose different challenges. If bevacizumab can be offered to only a limited number of patients, many clinicians favor those with the most aggressive disease, as these patients typically require a rapid response, require intensive therapy and do not have time to try less effective therapies first. This pragmatic approach in a resource-limited setting is supported by results of an analysis recently published by Bonotto and colleagues, who showed that failure to reach PFS of 6 months may be associated with reduced probability of benefit from subsequent lines of therapy compared with patients whose first-line therapy resulted in a PFS duration of ≥ 6 months [45]. This brings us to the fourth statement in the Delphi consensus.

The first challenge is the definition of 'high disease burden'. Prospective trials of treatment in such a specific population are lacking, not least because there is no agreed definition of 'high disease burden'. However, Arpino and colleagues conducted a systematic review of the literature to identify the factors most commonly associated with aggressive mBC [46]. The highest levels of evidence to support associations with worse OS were observed for visceral metastases, number of metastatic sites, disease-free interval, presence of circulating tumor cells, triple-negative disease and tumor grade.

In a meta-analysis of the three pivotal trials of first-line bevacizumab for mBC (albeit only one of them - E2100 - evaluated bevacizumab plus paclitaxel), treatment effect was explored in various populations of patients considered to have a poor prognosis [44]. The PFS HRs in subgroups of patients with triple-negative breast cancer (TNBC), visceral disease or ≥ 3 metastatic organ sites were 0.63, 0.66 and 0.64, respectively, remarkably similar to the PFS HR of 0.64 in the overall population. The absolute improvement in median PFS of 2-3 months with bevacizumab did not differ dramatically between subgroups, although in populations with very short PFS expectancy, the relative difference is greater and therefore perhaps of greater clinical importance.

In another retrospective analysis, this time based on the single-arm ATHENA study

evaluating first-line bevacizumab plus the investigator's choice of chemotherapy in the setting of routine oncology practice [47], Llombart-Cussac and colleagues developed a simple prognostic index for OS [48]. Using five clinical parameters, they were able to differentiate a population at high risk of death with a median OS of only 16 months on bevacizumab-containing therapy versus a population with a threefold lower risk of death and median OS of 39 months on bevacizumab-containing therapy. The risk factors contributing to this prognostic factor index were: liver metastases/ \geq 3 metastatic organ sites; triple-negative mBC; prior anthracycline and/or taxane therapy; Eastern Cooperative Oncology Group performance status 2 and/or prior analgesic and/or corticosteroid therapy; and a disease-free interval of ≤ 24 months. Subsequently, this prognostic factor index has been adopted in exploratory analyses of other bevacizumab mBC datasets and yielded supportive results. For example, in an exploratory analysis of the TURANDOT trial, which compared bevacizumab plus paclitaxel versus bevacizumab plus capecitabine [26], Brodowicz and colleagues concluded that in patients with triple-negative disease, a regimen of bevacizumab plus paclitaxel may be preferred to the bevacizumab plus capecitabine regimen based on the more favorable OS outcome despite shorter PFS [49].

5. Bevacizumab plus paclitaxel is an adequate first-line treatment option for TNBC

As already alluded to above, patients with TNBC – defined as negative estrogen receptor (ER), progesterone receptor (PgR) and HER2 status – generally have a poor prognosis. The population of patients with TNBC represents a heterogeneous subgroup, including at least six distinct subtypes (two basal-like, immunomodulatory, mesenchymal, mesenchymal stem-like and luminal androgen receptor) [50].

Considering the poor prognosis and limited treatment options for these patients, there has been particular interest in the role of bevacizumab in TNBC, not only in the metastatic setting but also as neoadjuvant and adjuvant therapy. Retrospective and subgroup analyses of Phase III trials demonstrated median PFS ranging from 8.8 to 10.6 months with bevacizumab plus paclitaxel [25,44,49]. In the subgroup of patients with TNBC in the ATHENA study, median time to disease progression with bevacizumab-containing therapy (not only paclitaxel) was 7.2 months [51]. These data led the panelists to conclude that bevacizumab plus paclitaxel is an adequate first-line treatment option for metastatic TNBC.

The importance of bevacizumab plus paclitaxel in triple-negative mBC is evidenced by the use of this regimen as the control arm in recent clinical trials, including the ongoing Dutch Triple-B randomized Phase IIB trial (NCT01898117) and a randomized Phase II trial of onartuzumab [52]. In this prospective trial in patients receiving first- or second-line therapy for TNBC, median PFS was 7.2 months with bevacizumab plus paclitaxel.

Although bevacizumab plus paclitaxel may be an adequate treatment option from the available choices, there is a need to find better treatments for patients with TNBC. Many ongoing clinical trials are evaluating new strategies for this difficult-to-treat population, with a particular focus on agents inhibiting PARP, such as veliparib, olaparib, niraparib and talazoparib [53,54]. Emerging data for immunotherapeutic approaches, such as the PD-1 inhibitor pembrolizumab [55] and PD-L1 inhibitor atezolizumab [56,57], highlight interest in this strategy for TNBC within an expansive range of tumor settings. The ongoing Phase III KEYNOTE-119 and IMpassion130 trials are evaluating pembrolizumab and atezolizumab, respectively, in metastatic TNBC. Other avenues of research include antiandrogenic therapies, MEK inhibition, PI3K inhibition, mTOR inhibition, heatshock protein 90 inhibition and histone deacetylase inhibition [53,54].

The landmark paper by Lehmann and colleagues [50], together with a simplified model that may allow easier translation into the clinic [58], provide new means of classifying TNBC that may potentially improve patient selection and guide therapeutic decisions. It is time to dissect this subgroup of breast cancer into more specific molecular entities, which will improve our ability to estimate prognosis and allow more efficacious treatment of our patients.

6. Bevacizumab plus paclitaxel is an appropriate first-line treatment option for hormone receptor (ER/PgR)-positive HER2-negative metastatic BC

For patients with hormone receptor-positive HER2-negative mBC, bevacizumab plus paclitaxel was also considered to be an appropriate first-line treatment option. This is perhaps not

surprising given that the benefit in TNBC in terms of HR was no different from that in the population with hormone receptor-positive disease or in the overall population [44]. However, additional data in this setting perhaps merit attention when considering the broader range of treatment options for these patients. First, according to the Advanced Breast Cancer 2 guidelines, the preferred treatment for patients with ER-positive HER2-negative mBC is endocrine therapy, even in the presence of visceral metastases [59,60]. Chemotherapy should be reserved for cases of rapidly progressing disease or proven endocrine resistance. Therefore the panelists agreed unanimously that bevacizumab plus paclitaxel is an appropriate firstline therapy for hormone receptor-positive HER2-negative mBC if chemotherapy is the only possible treatment option (e.g., in patients with hormone-refractory disease or those with visceral crisis).

Within the population of patients with hormone receptor-positive mBC, there exist different populations with different prognoses. For example, in the analysis of the ATHENA dataset described above, patients with hormone receptorpositive disease could be classified according to the prognostic factor index into those with one or no risk factors (median OS of 38.8 months), those with two risk factors (median OS of 23.9 months) and those with three or four risk factors (median OS of 17.4 months) [48]. Of note, in the subgroup with three or four risk factors, the OS expectancy was worse than in many of the patients with TNBC. Based on these observations, bevacizumab plus paclitaxel seems an appropriate treatment option for patients with hormone receptor-positive mBC, particularly those with more aggressive disease.

When a very similar prognostic factor index was applied to the TURANDOT dataset, patients with hormone receptor-positive mBC had more favorable PFS and ORR outcomes with the combination of bevacizumab plus paclitaxel compared with bevacizumab plus capecitabine [49]. However, interestingly, in the population of patients defined as having low-risk hormone receptor-positive mBC, those treated with bevacizumab plus capecitabine appeared to have a more favorable OS outcome. These exploratory analyses in a small, retrospectively defined subpopulation should have no bearing on clinical practice, especially in Italy where bevacizumab plus capecitabine is not available; however, they further illustrate the complexities of treatment decision-making for these patients.

It is also worth mentioning three trials in hormone receptor-positive mBC evaluating the combination of bevacizumab and endocrine therapy. The LEA and CALGB 40602 randomized Phase III trials both evaluated the addition of bevacizumab to endocrine therapy and in both, the primary end point was PFS [61,62]. In the LEA trial, there was no statistically significant improvement in PFS with the addition of bevacizumab to letrozole or fulvestrant. In the CALGB 40503 trial, the primary objective was met, demonstrating a significant PFS improvement with the addition of bevacizumab to letrozole (PFS: HR: 0.75, 95% CI: 0.59-0.96; p = 0.016; median 15.6 months with letrozole alone vs 20.2 months with letrozole plus bevacizumab). However, the authors noted a marked increase in grade 3/4 toxicities with bevacizumab-containing therapy. Finally, the AROBASE trial compared continued bevacizumab plus taxane versus a switch to bevacizumab plus exemestane after initial response to bevacizumab plus taxane induction therapy [63]. The trial was prematurely terminated because of the low probability of significantly improving PFS with the investigational arm.

7. In elderly patients, the use of bevacizumab is efficacious & safe as first-line treatment for HER2-negative mBC

While chronological age *per se* is not a reason to withhold a treatment, additional factors should be considered when selecting treatment for biologically elderly patients. Importantly, age in years is not always a reliable predictor of frailty and the importance of geriatric assessments is widely underestimated. Common comorbidities in older patients that may influence treatment choice include hypertension, hyperlipidemia, anemia, ischemic heart disease, diabetes, heart failure and chronic kidney disease [64].

To the best of our knowledge, no trials have evaluated bevacizumab specifically in older patients with mBC and experience is limited to exploratory and retrospective subgroup analyses. However, particularly in the ATHENA study, extensive analyses were undertaken to understand outcomes in patients aged \geq 70 years [65]. Both hypertension and proteinuria, known side effects of antiangiogenic therapies, were more common in patients aged \geq 70 years than in those younger than 70 years. However, in the older subgroup there was no excess of thromboembolic events, which are a particular concern in elderly patients. Overall, results from the ATHENA study provide no evidence for worse efficacy or tolerability in older compared with younger patients, although older age may influence chemotherapy choice [65].

8. Bevacizumab is also a valid treatment option in patients pretreated with (neo)adjuvant taxane in the early BC setting

Taxane rechallenge with a first-line taxane-containing regimen is generally a reasonable option for patients treated with taxane-containing therapy for early breast cancer, providing relapse has not occurred within 12 months of completing (neo)adjuvant taxane therapy [59,60]. In these patients, the addition of bevacizumab to first-line paclitaxel may be expected to improve efficacy based on the intent-to-treat results of Phase III trials of first-line bevacizumab [5,25,44]. The pooled analysis of three first-line trials (E2100, AVADO and RIBBON-1) suggested that taxane-pretreated patients may derive an OS benefit from the addition of bevacizumab but the lack of adjustment for multiplicity in these exploratory post hoc analyses is a major drawback when trying to interpret such findings [44].

• Real life

9. In the mBC setting, the data obtained in real-life studies confirm the efficacy of bevacizumab obtained in randomized clinical trials

There is no doubt that prospective randomized clinical trials remain the gold standard when determining the effect of a new treatment on patient outcomes. However, there is increasing interest in real-world data, in which outcomes are collected either prospectively or uniformly with a specific objective in mind or retrospectively from routine practice to address research questions in defined populations. Randomized clinical trials assess treatment effects within a specific carefully selected patient population and typically include strict eligibility criteria, rigorous monitoring for adherence to the protocol (including treatment administration and efficacy and safety assessment) and minimization of the risk of bias through randomization, stratification and usually double blinding. Although such trials undoubtedly provide the most rigorous evaluation of treatment effect, a limitation is their applicability to the populations of patients presenting in routine oncology practice, many of whom are ineligible for Phase III clinical trials.

Real-world data complement findings from randomized clinical trials and provide valuable information on the effectiveness of treatment, measuring the degree of clinical benefit in a realworld setting. Following release of results from the E2100 randomized Phase III trial evaluating first-line bevacizumab plus paclitaxel for mBC, the international ATHENA study was initiated. This large, open-label study evaluated first-line bevacizumab-containing therapy in a broader population of more than 2000 patients treated in 37 countries worldwide [47]. Importantly, the chemotherapy choice was at the investigator's discretion based on the clinical, disease and prior treatment characteristics of each patient. Approximately a third of patients (35%) received bevacizumab in combination with paclitaxel (17% in a weekly schedule, 13% in an every-3-week schedule; 6% other); 33% received bevacizumab with docetaxel, 10% with taxanebased combination regimens, and the remainder with nontaxane regimens, such as capecitabine or vinorelbine. Although the primary end point of ATHENA was safety, time to disease progression (interval between initiation of first-line therapy and recorded disease progression) was a predefined secondary end point. Median time to disease progression in ATHENA was 9.7 months in the entire population and 10.6 months in the subgroup of 325 patients treated with bevacizumab plus weekly paclitaxel [66]. Of specific interest to the panelists, median time to progression was 10.9 months in the subgroup of 278 patients treated in Italian clinical practice within the ATHENA trial [67].

Similar real-world studies have been undertaken in Hungary and Germany to evaluate the safety and efficacy of first-line bevacizumab plus paclitaxel in routine oncology practice. Median PFS was 9.3 months in 220 patients treated in the Hungarian AVAREG study [68] and 9.6 months in 865 patients treated in the German ML21165 study [69].

Turning to the second type of real-world data collection study mentioned above, very recently results were presented at the American Society of Clinical Oncology Annual Meeting from the Epidemio-Strategy-Medico-Economic database, assessing treatment outcomes in patients treated in routine oncology practice in France [70]. The study included 3426 patients starting firstline treatment with paclitaxel, with or without bevacizumab, between 2008 and 2013. OS was longer in patients treated with bevacizumab plus paclitaxel compared with paclitaxel alone (HR adjusted for the main prognostic variables: 0.67, 95% CI: 0.60–0.75; median 27.7 vs 19.8 months, respectively). Sensitivity analyses and supporting analyses adjusting for propensity score showed consistent results.

Taking into account differences in data collection and follow-up as well as the broader patient populations included in real-world studies, the panelists all agreed that these results are consistent with results from the prospective randomized Phase III trials of first-line bevacizumab plus paclitaxel in more strictly defined patient populations and suggest that results from Phase III evaluation can be transferred to routine oncology practice.

• Treatment duration

10. It is useful to continue the first-line treatment of mBC with bevacizumab until progression even after the discontinuation of chemotherapy

In the randomized Phase III trials of first-line bevacizumab-containing therapy for mBC, bevacizumab was generally continued until disease progression, unacceptable toxicity or withdrawal of consent. Retrospective analyses attempting to assess the impact of continued single-agent bevacizumab have suggested improved outcomes if bevacizumab is continued as maintenance therapy versus discontinued at the same time chemotherapy is stopped [66,71]. However, comparing outcomes in patients who received maintenance bevacizumab versus prematurely discontinuing bevacizumab is extremely challenging because duration of treatment and outcome are not independent: those patients remaining alive and progression-free (in the case of PFS) or alive (in the case of OS) for longer have the opportunity to receive bevacizumab for longer, whereas those who have early progression will discontinue bevacizumab earlier. In the absence of data specifically comparing the two strategies, the approach adopted in clinical trials and stated in the prescribing information should be followed (i.e., treatment with bevacizumab until disease progression or unacceptable toxicity).

Although the panelists agreed on the statement that bevacizumab should be continued until disease progression even after interrupting chemotherapy, one participant expressed concern that a prolonged duration of bevacizumab treatment may result in cumulative cardiovascular risk and potentially increase the severity of side effects such as hypertension and proteinuria. Investigators from at least three trials of bevacizumab plus paclitaxel have reported safety results specifically in subgroups of patients treated for ≥ 1 year. In the ATHENA study, 473 patients (21%) continued bevacizumab for ≥ 1 year (99 of whom continued bevacizumab for ≥ 2 years). The mean number of grade ≥ 3 adverse events was 1.26 per treatment-year in patients treated for ≥ 1 year versus 4.13 events per treatment-year in those treated for less than 1 year. Proteinuria was more common in the post 1-year treatment period suggesting that this side effect is cumulative; however, there was no evidence that first onset of hypertension was more common in later than earlier cycles of treatment [66]. In the single-arm JO19901 study evaluating first-line bevacizumab plus paclitaxel in Japanese patients, both hypertension and proteinuria occurred occasionally in later cycles [72]. In the single-arm German noninterventional ML21165 study, the incidence of grade ≥ 3 hypertension was increased with prolonged bevacizumab exposure but this effect was described as manageable by the investigators [69].

In the context of maintenance bevacizumab, results of the randomized Phase III IMELDA trial deserve mention. This trial compared maintenance bevacizumab alone versus maintenance bevacizumab plus capecitabine in patients achieving disease control with initial bevacizumab plus docetaxel therapy [73]. Unfortunately, recruitment to the trial was prematurely terminated following withdrawal of regulatory approval for the bevacizumab plus docetaxel combination. Despite this, the trial provided intriguing results. Patients randomized to receive capecitabine with maintenance bevacizumab had longer PFS (primary end point) and, remarkably, longer OS (secondary end point) compared with patients receiving maintenance bevacizumab alone. The practical application of these results is challenging, particularly in Italy where bevacizumab plus capecitabine is not available, yet the concept of induction therapy with a taxane plus bevacizumab followed by a switch to a more tolerable chemotherapy regimen with continued bevacizumab certainly generates interesting hypotheses. The authors concluded that these results might inform future maintenance trials and current first-line treatment strategies for HER2-negative mBC.

• Quality of life & safety

11. Combining bevacizumab with paclitaxel as first-line treatment for mBC does not negatively impact on QoL

As mentioned above, improving PFS may improve QoL [33], which is an important treatment goal in patients with mBC, particularly when improvement in OS is so elusive. QoL is generally measured by self-reported questionnaires, such as the Functional Assessment of Cancer Therapy-Breast (FACT-B) instrument. FACT-B, which is specifically designed to evaluate patient-reported outcomes in breast cancer, includes the 27-item FACT-General questionnaire assessing overall physical, functional, emotional and social well-being and the 9-item Breast Cancer Subscale, which focuses specifically on concerns of particular relevance to patients with breast cancer.

In the E2100 study, analyses of FACT-B results indicated that the addition of bevacizumab to paclitaxel was not associated with additional side-effect burden from the patients' perspective. Furthermore, breast cancer-specific concerns were reduced to a greater extent with bevacizumab plus paclitaxel than with paclitaxel alone [74]. In the TURANDOT trial, patient-reported outcomes were assessed using the European Organisation for Research and Treatment of Cancer Quality of Life Core Questionnaire 30 (EORTC QLQ-C30) at baseline, at each tumor assessment (every 12 weeks), and 28 days after discontinuation of study treatment. Analysis of mean global health status showed little change from baseline over time [26].

12. Bevacizumab has a well-defined safety profile

There can be little debate that bevacizumab has a well-defined safety profile after evaluation in a multitude of Phase III trials across a wide range of tumor types. Modest qualitative differences between tumor types have been noticed but overall, bevacizumab therapy is characterized by increased incidences of hypertension, proteinuria and low-grade bleeding. Specifically in mBC, the risk of hypertension, proteinuria, left ventricular dysfunction and hemorrhagic events is increased in patients receiving bevacizumab in combination with chemotherapy versus chemotherapy alone [75]. However, no statistically significant differences in the incidences of gastrointestinal perforation, vascular events or febrile neutropenia were observed. Adverse effects of bevacizumab have been analyzed in numerous meta-analyses in breast cancer and across tumor types: conclusions from these meta-analyses differ slightly according to the trial selection criteria but overall show quite similar results [44,75–77].

Conclusion

Despite the lack of consensus between regulatory authorities on either side of the Atlantic Ocean and even between European countries, the 31 Italian oncologists participating in this Delphi consensus reached a high level of agreement. These expert oncologists reached full (100%) consensus with regard to the efficacy of firstline bevacizumab plus paclitaxel, the clinical significance of the PFS benefit associated with this regimen despite the lack of OS benefit, translation of efficacy from clinical trials to the realworld setting and the well-defined safety profile of the regimen. Consensus was reached, but not unanimously, on topics related to the role of bevacizumab plus paclitaxel in specific populations of patients (those with high disease burden, TNBC, taxane-pretreated disease or advanced age), the continuation of bevacizumab until disease progression and the lack of negative impact of bevacizumab on QoL. Nevertheless, for these parameters the level of agreement with the statements exceeded 90% in all cases. Ultimately, in the absence of a clear predictive marker, treatment decisions regarding the administration of bevacizumab plus paclitaxel rest on the clinical judgement of the oncologist treating the patient, taking into consideration numerous factors related to patient, disease and medical history characteristics.

This consensus represents an Italian perspective. We believe that the consensus reached here could provide valuable insight for the development of practice guidelines at institutional or regional levels in Italy. It would be interesting to know whether similar consensus could be reached by surveying a pan-European panel of oncologists, which would be expected to reflect differences in treatment practice, local guidelines, regulatory status and reimbursement issues.

Ongoing and future research in mBC is focusing on newer strategies targeting a range of other pathways, including PI3K and MEK, and immunotherapy as well as vaccination strategies. Increasingly a more personalized approach to treatment and a more targeted approach to drug development are being embraced.

Future perspective

In the next few years, the first-line treatment of HER2-negative mBC is expected to change with new advances in our knowledge of tumor biology. Molecularly driven treatment could be of benefit if the right target is identified. Accordingly, for each line of treatment it is desirable to identify predictive factors that may influence the therapeutic strategy. Parsimonious use of chemotherapy is of value as well as integration of chemotherapy into regimens including different types of biologic agent (e.g., antiangiogenic agents, immunotherapy).

Financial & competing interests disclosure

F Puglisi was a member of the Steering Committee for the Roche-sponsored TANIA trial (ClinicalTrials.gov Identifier: NCT01250379) and has received honoraria for speaker engagements and acted as a consultant for Roche. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

Medical writing support for this manuscript, provided by Jennifer Kelly, was funded by an unrestricted educational grant by Roche Italy.

EXECUTIVE SUMMARY

- Bevacizumab plus paclitaxel is an effective first-line treatment option for HER2-negative metastatic breast cancer, providing a clinically meaningful benefit over paclitaxel alone.
- No subgroup deriving a particularly large or small benefit from bevacizumab has been identified; therefore patient selection for bevacizumab is typically based on clinical need and availability of treatments.
- Results from real-world studies, in particular from almost 3500 patients treated in routine oncology practice in France, demonstrate the effectiveness of first-line bevacizumab plus paclitaxel for HER2-negative metastatic breast cancer.
- In the absence of robust data comparing continuing versus discontinuing bevacizumab after cessation of chemotherapy, continuation of maintenance bevacizumab (ideally with capecitabine) until disease progression is advisable.
- The safety profile of bevacizumab has been extensively described; the characteristic adverse events of hypertension
 and proteinuria can be managed relatively easily, and side effects associated with bevacizumab do not appear to have
 a detrimental effect on quality of life.

References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- Dawood S, Shaikh AJ, Buchholz TA *et al.* The use of bevacizumab among women with metastatic breast cancer: a survey on clinical practice and the ongoing controversy. *Cancer* 118(11), 2780–2786 (2012).
- 2 Montero AJ, Escobar M, Lopes G, Glück S, Vogel C. Bevacizumab in the treatment of metastatic breast cancer: friend or foe? *Curr. Oncol. Rep.* 14(1), 1–11 (2012).
- 3 Cortés J, Calvo E, González-Martín A *et al.* Progress against solid tumors in danger: the metastatic breast cancer example. *J. Clin. Oncol.* 30(28), 3444–3447 (2012).
- 4 Lyman GH, Burstein HJ, Buzdar AU, D'Agostino R, Ellis PA. Making genuine progress against metastatic breast cancer. *J. Clin. Oncol.* 30(28), 3448–3451 (2012).
- 5 Miller K, Wang M, Gralow J *et al.* Paclitaxel plus bevacizumab versus paclitaxel alone for metastatic breast cancer. *N. Engl. J. Med.* 357(26), 2666–2676 (2007).

- 6 Robert NJ, Diéras V, Glaspy J et al. RIBBON-1: randomized, double-blind, placebo-controlled, Phase III trial of chemotherapy with or without bevacizumab for first-line treatment of human epidermal growth factor receptor 2-negative, locally recurrent or metastatic breast cancer. J. Clin. Oncol. 29(10), 1252–1260 (2011).
- 7 Miles DW, Chan A, Dirix LY *et al.* Phase III study of bevacizumab plus docetaxel compared with placebo plus docetaxel for the first-line treatment of human epidermal growth factor receptor 2-negative metastatic breast cancer. *J. Clin. Oncol.* 28(20), 3239–3247 (2010).
- Bear HD, Tang G, Rastogi P *et al.* Bevacizumab added to neoadjuvant chemotherapy for breast cancer. *N. Engl. J. Med.* 366, 310–320 (2012).
- 9 von Minckwitz G, Eidtmann H, Rezai M et al. Arbeitsgemeinschaft Gynäkologische Onkologie–Breast Study Groups. Neoadjuvant chemotherapy and bevacizumab for HER2-negative breast cancer. N. Engl. J. Med. 366(4), 299–309 (2012).

- Sikov WM, Berry DA, Perou CM *et al.* Impact of the addition of carboplatin and/or bevacizumab to neoadjuvant once-per-week paclitaxel followed by dose-dense doxorubicin and cyclophosphamide on pathologic complete response rates in stage II to III triple-negative breast cancer: CALGB 40603 (Alliance). *J. Clin. Oncol.* 33(1), 13–21 (2015).
- 11 Earl HM, Hiller L, Dunn JA et al. ARTemis Investigators. Efficacy of neoadjuvant bevacizumab added to docetaxel followed by fluorouracil, epirubicin, and cyclophosphamide, for women with HER2-negative early breast cancer (ARTemis): an open-label, randomised, Phase 3 trial. *Lancet Oncol.* 16(6), 656–666 (2015).
- Andre F, Deluche E, Bonnefoi H. Bevacizumab: the phoenix of breast oncology? *Lancet Oncol.* 16(6), 600–601 (2015).
- Bear HD, Tang G, Rastogi P *et al.* Neoadjuvant plus adjuvant bevacizumab in early breast cancer (NSABP B-40

PERSPECTIVE Puglisi, Bisagni, Ciccarese et al.

[NRG Oncology]): secondary outcomes of a Phase 3, randomised controlled trial. *Lancet Oncol.* 16(9), 1037–1048 (2015).

- 14 von Minckwitz G, Loibl S, Untch M et al. GBG/AGO-B study groups. Survival after neoadjuvant chemotherapy with or without bevacizumab or everolimus for HER2-negative primary breast cancer (GBG 44-GeparQuinto). Ann. Oncol. 25(12), 2363–2372 (2014).
- 15 Earl H, Hiller L, Dunn J et al. Disease-free (DFS) and overall survival (OS) at 3.4 years (yrs) for neoadjuvant bevacizumab (Bev) added to docetaxel followed by fluorouracil, epirubicin and cyclophosphamide (D-FEC), for women with HER2 negative early breast cancer: the ARTemis trial. J. Clin. Oncol. 34(Suppl.), Abstract 1014 (2016).
- 16 Gill FJ, Leslie GD, Grech C, Latour JM. Using a web-based survey tool to undertake a Delphi study: application for nurse education research. *Nurse Educ. Today* 33(11), 1322–1328 (2013).
- 17 Nyborg G, Straand J, Klovning A, Brekke M. The Norwegian General Practice–Nursing Home criteria (NORGEP-NH) for potentially inappropriate medication use: a web-based Delphi study. *Scand. J. Prim. Health Care* 33(2), 134–141 (2015).
- 18 Garvelink MM, Ter Kuile MM, Louwé LA, Hilders CG, Stiggelbout AM. A Delphi consensus study among patients and clinicians in the Netherlands on the procedure of informing young breast cancer patients about fertility preservation. *Acta Oncol.* 51(8), 1062–1069 (2012).
- 19 Girolomoni G, Altomare G, Ayala F *et al.* Differential management of mild-to-severe psoriasis with biologic drugs: an Italian Delphi consensus expert panel. *J. Dermatolog. Treat.* 26(2), 128–133 (2015).
- 20 Thomas SL, Wakerman J, Humphreys JS. Ensuring equity of access to primary health care in rural and remote Australia – what core services should be locally available? *Int. J. Equity Health* 14, 111 (2015).
- 21 Terrone C, Berruti A, Papotti M, Vavassori V, Sciarra A. Management of patients with castration-resistant prostate cancer (CRPC): results of an Italian survey using the Delphi method. *Tumori* doi:10.5301/tj.5000389 (2015) (Epub ahead of print).
- 22 Mohile SG, Velarde C, Hurria A et al. Geriatric assessment-guided care processes for older adults: a Delphi consensus of geriatric oncology experts. J. Natl Compr. Canc. Netw. 13(9), 1120–1130 (2015).
- 23 Agenzia Italiana del Farmaco, 2014. www.agenziafarmaco.gov.it

- 24 Gray R, Bhattacharya S, Bowden C, Miller K, Comis RL. Independent review of E2100: a Phase III trial of bevacizumab plus paclitaxel versus paclitaxel in women with metastatic breast cancer. J. Clin. Oncol. 27(30), 4966–4972 (2009).
- 25 Miles D, Cameron D, Bondarenko I et al. First results from the double-blind placebo (PL)-controlled randomised Phase III MERIDIAN trial prospectively evaluating plasma (p)VEGF-A in patients (pts) receiving first-line paclitaxel (PAC) ± bevacizumab (BV) for HER2-negative metastatic breast cancer (mBC). Eur. J. Cancer 51(Suppl. 3), S287 (2015) (Abstract 1866).
- This is the first randomized Phase III trial prospectively evaluating the potential of plasma VEGF-A as a predictive biomarker for bevacizumab efficacy, based on a suggested role in several retrospective analyses.
- 26 Lang I, Brodowicz T, Ryvo L et al. Central European Cooperative Oncology Group. Bevacizumab plus paclitaxel versus bevacizumab plus capecitabine as first-line treatment for HER2-negative metastatic breast cancer: interim efficacy results of the randomised, open-label, non-inferiority, Phase 3 TURANDOT trial. Lancet Oncol. 14(2), 125–133 (2013).
- 27 Rugo HS, Barry WT, Moreno-Aspitia A et al. Randomized Phase III trial of paclitaxel once per week compared with nanoparticle albumin-bound nab-paclitaxel once per week or ixabepilone with bevacizumab as first-line chemotherapy for locally recurrent or metastatic breast cancer: CALGB 40502/ NCCTG N063H (Alliance). J. Clin. Oncol. 33(21), 2361–2369 (2015).
- 28 Lück HJ, Lübbe K, Reinisch M et al. Phase III study on efficacy of taxanes plus bevacizumab with or without capecitabine as first-line chemotherapy in metastatic breast cancer. Breast Cancer Res. Treat. 149(1), 141–149 (2015).
- 29 Bruzzi P, Del Mastro L, Sormani MP et al. Objective response to chemotherapy as a potential surrogate end point of survival in metastatic breast cancer patients. J. Clin. Oncol. 23, 5117–5125 (2005).
- 30 Burzykowski T, Buyse M, Piccart-Gebhart MJ et al. Evaluation of tumor response, disease control, progression-free survival, and time to progression as potential surrogate end points in metastatic breast cancer. J. Clin. Oncol. 26(12), 1987–1992 (2008).
- 31 Ellis LM, Bernstein DS, Voest EE et al. American Society of Clinical Oncology perspective: raising the bar for clinical trials

by defining clinically meaningful outcomes. J. Clin. Oncol. 32(12), 1277–1280 (2014).

- 32 Cherny NI, Sullivan R, Dafni U et al. A standardised, generic, validated approach to stratify the magnitude of clinical benefit that can be anticipated from anti-cancer therapies: the European Society for Medical Oncology Magnitude of Clinical Benefit Scale (ESMO-MCBS). Ann. Oncol. 26(8), 1547–1573 (2015).
- 33 Hurvitz SA, Lalla D, Crosby RD, Mathias SD. Use of the metastatic breast cancer progression (MBC-P) questionnaire to assess the value of progression-free survival for women with metastatic breast cancer. *Breast Cancer Res. Treat.* 142(3), 603–609 (2013).
- 34 Hurvitz SA. Evolving options for the treatment of metastatic breast cancer: progression-free survival as an endpoint. *Cancer Treat. Rev.* 37(7), 495–504 (2011).
- 35 Booth CM, Eisenhauer EA. Progression-free survival: meaningful or simply measurable? *J. Clin. Oncol.* 30(10), 1030–1033 (2012).
- 36 Broglio KR, Berry DA. Detecting an overall survival benefit that is derived from progression-free survival. J. Natl Cancer Inst. 101, 1642–1649 (2009).
- 37 Adunlin G, Cyrus JW, Dranitsaris G. Correlation between progression-free survival and overall survival in metastatic breast cancer patients receiving anthracyclines, taxanes, or targeted therapies: a trial-level meta-analysis. *Breast Cancer Res. Treat.* 154, 591–608 (2015).
- 38 Raphael J, Verma S. Overall survival (OS) endpoint: an incomplete evaluation of metastatic breast cancer (MBC) treatment outcome. *Breast Cancer Res. Treat.* 150(3), 473–478 (2015).
- 39 Paez-Ribes M, Man S, Xu P, Kerbel RS. Potential proinvasive or metastatic effects of preclinical antiangiogenic therapy are prevented by concurrent chemotherapy. *Clin. Cancer Res.* 21(24), 5488–5498 (2015).
- 40 Miles D, Harbeck N, Escudier B *et al.* Disease course patterns after discontinuation of bevacizumab: pooled analysis of randomized Phase III trials. *J. Clin. Oncol.* 29, 83–88 (2011).
- 41 Chen L, Linden HM, Anderson BO, Li CI. Trends in 5-year survival rates among breast cancer patients by hormone receptor status and stage. *Breast Cancer Res. Treat.* 147(3), 609–616 (2014).
- 42 Puglisi F, Rea D, Kroes MA, Pronzato P. Second-line single-agent chemotherapy in human epidermal growth factor receptor 2-negative metastatic breast cancer: a

systematic review. *Cancer Treat. Rev.* 43, 36–49 (2016).

- 43 Lambrechts D, Lenz HJ, de Haas S *et al.* Markers of response for the antiangiogenic agent bevacizumab. *J. Clin. Oncol.* 31, 1219–1230 (2012).
- 44 Miles DW, Diéras V, Cortés J, Duenne AA, Yi J, O'Shaughnessy J. First-line bevacizumab in combination with chemotherapy for HER2-negative metastatic breast cancer: pooled and subgroup analyses of data from 2447 patients. *Ann. Oncol.* 24(11), 2773–2780 (2013).
- This pooled meta-analysis of individual patient data from the three pivotal trials of first-line bevacizumab-containing therapy for metastatic breast cancer (mBC) demonstrated benefit in patients with poor-prognosis disease and a short overall survival expectancy.
- 45 Bonotto M, Gerratana L, Iacono D *et al.* Treatment of metastatic breast cancer in a real-world scenario: is progression-free survival with first line predictive of benefit from second and later lines? *Oncologist* 20, 719–724 (2015).
- 46 Arpino G, Milano M, De Placido S. Features of aggressive breast cancer. *Breast* 24, 594–600 (2015).
- 47 Smith IE, Pierga JY, Biganzoli L *et al.* ATHENA Study Group. First-line bevacizumab plus taxane-based chemotherapy for locally recurrent or metastatic breast cancer: safety and efficacy in an open-label study in 2,251 patients. *Ann. Oncol.* 22(3), 595–602 (2011).
- 48 Llombart-Cussac A, Pivot X, Biganzoli L et al. A prognostic factor index for overall survival in patients receiving first-line chemotherapy for HER2-negative advanced breast cancer: an analysis of the ATHENA trial. Breast 23(5), 656–662 (2014).
- •• Using a dataset of more than 2000 patients treated with first-line bevacizumabcontaining therapy for HER2-negative mBC, the authors developed a simple prognostic factor index for overall survival that enabled identification of a population of patients with a threefold shorter overall survival expectancy compared with patients with a better prognosis.
- 49 Brodowicz T, Lang I, Kahan Z et al. Selecting first-line bevacizumab-containing therapy for advanced breast cancer: TURANDOT risk factor analyses. Br. J. Cancer 111(11), 2051–2057 (2014).
- 50 Lehmann BD, Bauer JA, Chen X *et al.* Identification of human triple-negative breast

cancer subtypes and preclinical models for selection of targeted therapies. *J. Clin. Invest.* 121(7), 2750–2767 (2011).

- 51 Thomssen C, Pierga JY, Pritchard KI et al. First-line bevacizumab-containing therapy for triple-negative breast cancer: analysis of 585 patients treated in the ATHENA study. Oncology 82(4), 218–227 (2012).
- 52 Diéras V, Campone M, Yardley DA et al. Randomized, Phase II, placebo-controlled trial of onartuzumab and/or bevacizumab in combination with weekly paclitaxel in patients with metastatic triple-negative breast cancer. Ann. Oncol. 26(9), 1904–1910 (2015).
- 53 Marmé F, Schneeweiss A. Targeted therapies in triple-negative breast cancer. *Breast Care* (*Basel*) 10(3), 159–166 (2015).
- 54 Bianchini G, Balko JM, Mayer IA, Sanders ME, Gianni L. Triple-negative breast cancer: challenges and opportunities of a heterogeneous disease. *Nat. Rev. Clin. Oncol.* doi:10.1038/nrclinonc.2016.66 (2016) (Epub ahead of print).
- Describes the molecular heterogeneity of triple-negative breast cancer and provides a comprehensive overview of the ongoing research efforts to identify the most effective targeted treatment options for these patients.
- 55 Nanda R, Chow LQ, Dees EC et al. Pembrolizumab in patients with advanced triple-negative breast cancer: Phase Ib KEYNOTE-012 study. J. Clin. Oncol. doi:10.1200/JCO.64.8931 (2016) (Epub ahead of print).
- 56 Emens LA, Braiteh FS, Cassier P et al. Inhibition of PD-L1 by MPDL3280A leads to clinical activity in patients with metastatic triple-negative breast cancer (TNBC). Cancer Res. 75(15 Suppl.), Abstract 2859 (2015).
- 57 Adams S, Robinson Diamond J, Hamilton EA et al. Phase Ib trial of atezolizumab in combination with nab-paclitaxel in patients with metastatic triple-negative breast cancer (mTNBC). J. Clin. Oncol. 34(Suppl.), Abstract 1009 (2016).
- 58 Ring BZ, Hout DR, Morris SW *et al.* Generation of an algorithm based on minimal gene sets to clinically subtype triple negative breast cancer patients. *BMC Cancer* 16(1), 143 (2016). Erratum in: *BMC Cancer* 16(1), 275 (2016).
- 59 Cardoso F, Costa A, Norton L et al. ESO-ESMO 2nd international consensus guidelines for advanced breast cancer (ABC2). Ann. Oncol. 25(10), 1871–1888 (2014).
- 60 Cardoso F, Costa A, Norton L *et al.* ESO-ESMO 2nd international consensus

guidelines for advanced breast cancer (ABC2). *Breast.* 23(5), 489–502 (2014).

- 61 Martín M, Loibl S, von Minckwitz G *et al.* Phase III trial evaluating the addition of bevacizumab to endocrine therapy as first-line treatment for advanced breast cancer: the letrozole/fulvestrant and avastin (LEA) study. *J. Clin. Oncol.* 33(9), 1045–1052 (2015).
- 62 Dickler MN, Barry WT, Cirrincione CT et al. Phase III trial evaluating letrozole as first-line endocrine therapy with or without bevacizumab for the treatment of postmenopausal women with hormone receptor-positive advanced-stage breast cancer: CALGB 40503 (Alliance). J. Clin. Oncol. doi:10.1200/JCO.66.1595 (2016) (Epub ahead of print).
- 63 Trédan O, Follana P, Moullet I *et al.* A Phase III trial of exemestane plus bevacizumab maintenance therapy in patients with metastatic breast cancer after first-line taxane and bevacizumab: a GINECO group study. *Ann. Oncol.* 27(6), 1020–1029 (2016).
- 64 Partridge AH, Rumble RB, Carey LA *et al.* Chemotherapy and targeted therapy for women with human epidermal growth factor receptor 2-negative (or unknown) advanced breast cancer: American Society of Clinical Oncology Clinical Practice Guideline. *J. Clin. Oncol.* 32(29), 3307–3329 (2014).
- 65 Biganzoli L, Di Vincenzo E, Jiang Z et al.
 First-line bevacizumab-containing therapy for breast cancer: results in patients aged
 ≥70 years treated in the ATHENA study.
 Ann. Oncol. 23(1), 111–118 (2012).
- 66 Smith I, Pierga JY, Biganzoli L *et al.* Final overall survival results and effect of prolonged (≥ 1 year) first-line bevacizumab-containing therapy for metastatic breast cancer in the ATHENA trial. *Breast Cancer Res. Treat.* 130(1), 133–143 (2011).
- 67 Sanna G, Fabi A, Crivellari D *et al.* Safety and efficacy of bevacizumab in combination with first-line chemotherapy in advanced breast cancer: data from the Italian cohort of the ATHENA trial. *Tumori* 100(4), 432–438 (2014).
- 68 Dank M, Budi L, Piko B et al. First-line bevacizumab-paclitaxel in 220 patients with metastatic breast cancer: results from the AVAREG study. Anticancer Res. 34(3), 1275–1280 (2014).
- 69 Schneeweiss A, Förster F, Tesch H et al. First-line bevacizumab-containing therapy for HER2-negative metastatic breast cancer: final results from a prospective German study. Anticancer Res. 36(3), 967–974 (2016).
- 70 Delaloge S, Perol D, Brain E *et al.* Overall survival of patients with HER2-negative

PERSPECTIVE Puglisi, Bisagni, Ciccarese et al.

metastatic breast cancer treated with a first-line paclitaxel with or without bevacizumab in real-life setting: results of a multicenter national observational study. Presented at: *American Society of Clinical Oncology.* Chicago, IL, USA, 3–7 June 2016 (Abstract 1013).

- •• This real-world study suggests that in patients receiving first-line paclitaxel for HER2-negative mBC, the addition of bevacizumab is associated with significantly improved overall survival.
- 71 Redondo A, Martinez V, Zamora P *et al.* Continuation of bevacizumab and addition of hormone therapy following weekly paclitaxel therapy in HER2-negative metastatic breast cancer. *Onco Targets Ther.* 7, 2175–2181 (2014).
- 72 Aogi K, Masuda N, Ohno S *et al.* First-line bevacizumab in combination with weekly paclitaxel for metastatic breast cancer: efficacy and safety results from a large, open-label, single-arm Japanese study. *Breast Cancer Res. Treat.* 129(3), 829–838 (2011).
- 73 Gligorov J, Doval D, Bines J et al. Maintenance capecitabine and bevacizumab versus bevacizumab alone after initial first-line bevacizumab and docetaxel for patients with HER2-negative metastatic breast cancer (IMELDA): a randomised, open-label, Phase 3 trial. *Lancet Oncol.* 15(12), 1351–1360 (2014).
- 74 Cella D, Wang M, Wagner L, Miller K. Survival-adjusted health-related quality of life (HRQL) among patients with metastatic breast cancer receiving paclitaxel plus bevacizumab versus paclitaxel alone: results

from Eastern Cooperative Oncology Group Study 2100 (E2100). *Breast Cancer Res. Treat.* 130(3), 855–861 (2011).

- 75 Cortes J, Calvo V, Ramírez-Merino N et al. Adverse events risk associated with bevacizumab addition to breast cancer chemotherapy: a meta-analysis. Ann. Oncol. 23(5), 1130–1137 (2012).
- 76 Li Q, Yan H, Zhao P, Yang Y, Cao B. Efficacy and safety of bevacizumab combined with chemotherapy for managing metastatic breast cancer: a meta-analysis of randomized controlled trials. *Sci. Rep.* 5, 15746 (2015).
- 77 Fang Y, Qu X, Cheng B et al. The efficacy and safety of bevacizumab combined with chemotherapy in treatment of HER2-negative metastatic breast cancer: a meta-analysis based on published Phase III trials. *Tumour Biol.* 36(3), 1933–1941 (2015).